Author’s response to reviews

Title: Bacteremia in Hospitalized Patients With Human Immunodeficiency Virus: A Prospective, Cohort Study

Authors:

Bekele Afessa (schwartz.roberta@mayo.edu)
Ian Morales (schwartz.roberta@mayo.edu)
Bethany Weaver (schwartz.roberta@mayo.edu)

Version: 5 Date: 5 Sep 2001

Ruth King, Editorial Assistant
BMC Infectious Diseases
Middlesex House
34-42 Cleveland Street
London W1T 4LB

Dear Ms. King:

The enclosed 2nd revised manuscript

Afessa B, Morales I, Weaver B. Bacteremia in Hospitalized Patients With Human Immunodeficiency Virus: A Prospective, Cohort Study

is resubmitted for your consideration for publication in BMC Infectious Diseases.

As you requested, all authors’ names have been linked to their corresponding affiliation using superscript numerals; we have included borders within the tables to ensure that each table is duplicated properly in the web version; and we have added a Competing Interests statement.

To facilitate prompt handling, we ask that you direct correspondence and proof to the authors c/o Section of Scientific Publications, Mayo Clinic, 200 First Street SW, Rochester, MN 55905. Our telephone number is 507-284-3335, our fax number is 507-284-2107, and our E-mail address is schwartz.roberta@mayo.edu.
Sincerely yours,

Mrs. Roberta Schwartz
Production Editor

RS:ml
Enc.: manuscript